Author’s response to reviews

Title: Fatalism in breast cancer and performing mammography on women with or without a family history of breast cancer

Authors:

Maryam Molaei_Zardanjani (m_molaei@nm.mui.ac.ir)
Mitra Savabi Esfahani (m_savabi@nm.mui.ac.ir)
Fariba Taleghani (Taleghani@nm.mui.ac.ir)

Version: 1 Date: 04 May 2019

Author’s response to reviews:

Comment 1:
You should have your manuscript reviewed by someone who is fluent in English.

Answer1:
Changes have been done in total manuscript.

Comment 2:
For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must confirm this under this sub-heading and also provide their reasons.

Answer2:
We have provided more detail in page 6.

Comment 3:
For further guidance on how to format this section, please refer to BioMed Central's editorial policies page (see links below).
Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements

Answer:

We have mentioned in pages 5-6.